Title: Rizatriptan versus rizatriptan plus rofecoxib versus rizatriptan plus tolfenamic acid in the acute treatment of migraine

Authors:

Abouch V Krymchantowski (abouchkrym@globo.com)
Marcelo E Bigal (marcbigal@aol.com)

Version: 3 Date: 25 June 2004

Author's response to reviews:

Corrections on the affiliations, removal of the author's qualifications, inclusion of the countries names, removal of keywords, revision of the capitalisation - DONE Rewriting of the abstract with the removal of unnecessary sections - DONE Revision of manuscript details, square brackets in reference numbers, inclusion of competing interests, author's contributions and acknowledgements - DONE Upload of the new abstract - Done